\documentclass[12pt,draft]{amsart}
\setlength{\topmargin}{-1.25in}
%\setlength{\oddsidemargin}{0in}
%\setlength{\textwidth}{6.5in}
\setlength{\textheight}{11in}
\setlength{\parindent}{0in}
\pagestyle{empty}

\begin{document}
\thispagestyle{empty}

\ \vfill

\begin{center}
{\Large \begin{bf}
Project Based Math 112, Fall 2001\\
Homework \#3\\
due Monday, September 24\\[.25in]
\end{bf}}
\end{center}


\large{
Read sections \S 6.2, \S 5.6 and \S 5.7 in the book. \\

Do all problems.  Those marked with a * will be graded. \\

\S 6.2: 2, 8*, 12, 13, 23, 26*, 30*, 31, 46*\\[.2cm]
\S 5.6: 2, 9, 12, 18*, 21, 22*, 26*, 33, 34*\\[.2cm]
\S 5.7: 2*, 6, 10, 14*, 16a,b, 18*, 27, 30\\[.2cm]
}

\vfill\vfill

\begin{center}
{\Large \begin{bf}
Project Based Math 112, Fall 2001\\
Homework \#3\\
due Monday, September 24\\[.25in]
\end{bf}}
\end{center}


\large{
Read sections \S 6.2, \S 5.6 and \S 5.7 in the book. \\

Do all problems.  Those marked with a * will be graded. \\

\S 6.2: 2, 8*, 12, 13, 23, 26*, 30*, 31, 46*\\[.2cm]
\S 5.6: 2, 9, 12, 18*, 21, 22*, 26*, 33, 34*\\[.2cm]
\S 5.7: 2*, 6, 10, 14*, 16a,b, 18*, 27, 30\\[.2cm]
}

\vfill

\end{document}
